Author’s response to reviews

Title: Outcomes after prostate cancer in France: treatments, side-effects and mortality at two years

Authors:

Philippe Tuppin (philippe.tuppin@cnamts.fr)
Solene Samson (Solene.samson@cnamts.fr)
Anne Fagot-Campagna (anne.fagot@cnamts.fr)
Bertrand Lukacs (bertrand.lukac@gmail.com)
Francois Alla (francois.alla@cnamts.fr)
Fred Paccaud (fred.paccaud@chuv.ch)
Jean-Christophe Thalabard (jean-christophe.thalabard@htd.aphp.fr)
Eric Vicaut (eric.vicaut@lrb.aphp.fr)
Michel Vidaud (michel.vidaud@parisdescartes.fr)
Bertrand Millat (b-millat@chu-montpellier.fr)

Version: 2

Date: 13 February 2014

Author’s response to reviews:

For this study, all data were gathered from the French National Health Insurance Information System (SNII RAM). Access to the data and analysis are granted by decree only for three years plus the current year with also an authorization by the French Data protection Authority (Commission nationale de l'informatique et des libertés, CNIL). Consequently, ethic approval is not mandatory. This point was added in the beginning of the methods section.

There was no specific funding for this study conducted by the employees of the French national general health insurance scheme in charge of the French National Health Insurance Information System (SNII RAM). It was added in competing interests chapter.